**Author's response to reviews**

**Title:** The Effect of Warm-up, Static Stretching and Dynamic Stretching on Hamstring Flexibility

**Authors:**

- Kieran O'Sullivan (kieran.osullivan@ul.ie)
- Elaine Murray (elainemurr@gmail.com)
- David Sainsbury (david.sainsbury@ul.ie)

**Version:** 2  **Date:** 17 December 2008

**Author's response to reviews:**

I am just submitting a revision with a clinical trials register number at the bottom of the abstract. This number was obtained from the Australia New Zealand Clinical trials register.